Author's response to reviews

**Title:** A Comparison of a Social Support Physical Activity Intervention in Weight Management among Post-partum Latinas

**Authors:**

Colleen Keller (Colleen.Keller@asu.edu)
Barbara Ainsworth (Barbara.Ainsworth@asu.edu)
Kathryn Records (Kathie.Records@asu.edu)
Michael Todd (Mike.Todd@asu.edu)
Michael Belyea (Michael.Belyea@asu.edu)
Sonia Vega-López (Sonia.Vega.Lopez@asu.edu)
Paska Permana (Paska.Permana@va.gov)
Dean Coonrod (Dean_Coonrod@dmgaz.org)

**Version:** 2 **Date:** 26 July 2013

**Author's response to reviews:** see over
July 26, 2013

Natalie Pafitis,
Executive Editor
BMC Public Health
BioMed Central
Floor 6, 236 Gray’s Inn Road
WC1X 8HB

RE: 2086063383102648 – A Comparison of a Social Support Physical Activity Intervention in Weight Management among Post-partum Latinas

Dear Dr. Pafitis,

We are submitting this manuscript “A Comparison of a Social Support Physical Activity Intervention in Weight Management among Post-partum Latinas” for your editorial consideration. We acknowledge the feedback offered by the journal. These comments focused on 3 primary areas a) Abstract format according to guidelines, b) Trial registration number, and c) Ethics committee specific name. We formatted the abstract according to guidelines. The Madres para la Salud [Madres for Health] trial registration number for ClinicalTrials.gov is: NCT01908959. The Office of Research Integrity and Assurance: Internal Review Board at Arizona State University is the ethics committee that granted approval for this controlled trial.

We look forward to your consideration.

Sincerely,

Colleen Keller, PhD, FAHA, FAAN
Regents’ Professor
Foundation Professor in Women’s Health and
Director, Center for Healthy Outcomes in Aging
Co-Director, T32 Transdisciplinary Training in Health Disparities Science